Author’s response to reviews

Title: Identification of a novel S6K1 inhibitor, rosmarinic acid methyl ester, for treating cisplatin-resistant cervical cancer

Authors:
Ki Hong NAM (nam6422@hanmail.net)
Sang Ah Yi (angelna1023@hanmail.net)
Gibeom Nam (gobohoboho@naver.com)
Jae Sung Noh (njs2429@naver.com)
Jong Woo Park (jongwoopark17@gmail.com)
Min Gyu Lee (sinnamonic@gmail.com)
Jee Hun Park (croi89@naver.com)
Hwamok Oh (moogkho@hanmail.net)
Jieun Lee (jjiny75@gmail.com)
Kang Ro Lee (krlee@skku.edu)
Hyun-Ju Park (hyunju85@skku.edu)
Jaecheol Lee (jaecheol@skku.edu)
Jeung-Whan Han (jhhan551@skku.edu)

Version: 2 Date: 26 May 2019

Author’s response to reviews:

We appreciate the editor for giving the chance of further revision. Below is our response to the point raised for minor revision of our manuscript.

Editor Comments:

Thank you for allowing us to review your manuscript, "Identification of a novel S6K1 inhibitor, rosmarinic acid methyl ester, for treating cisplatin-resistant cervical cancer". The revised manuscript...
reflects the appropriate response to the reviewer’s comments. However, some of western blot results displayed on the main and suppl. Figures are missing internal controls such as beta-actin. It should be needed.

- As the editor pointed out, we missed displaying the internal controls in Figure 6a and Supplementary Figure S4. Thus, we performed stripping with the membranes and detected actin as a loading control. The data are displayed in revised version of Figure 6a and Supplementary Figure S4.